Author’s response to reviews

Title: Characteristics of pncA Mutations in Multidrug-resistant Tuberculosis in Taiwan

Authors:

Yu-Chi Chiu (newchiuyuchi@gmail.com)
Shiang-Fen Huang (sfhuang6@vghtpe.gov.tw)
Kwok-Woon Yu (kwyu@vghtpe.gov.tw)
Yu-Chin Lee (leeyc@vghtpe.gov.tw)
Jia-Yih Feng (jyfeng@vghtpe.gov.tw)
Wei-Juin Su (wjsu@vghtpe.gov.tw)

Version: 5 Date: 31 August 2011

Author’s response to reviews:

Dr. Roxane Rajabi
Editor, BMC Infectious Disease
Re: Characteristics of pncA Mutations in Multidrug-resistant Tuberculosis in Taiwan (MS ID: 1275184982533499 version 4)

Dear Dr. Rajabi

The manuscript has been revised in accordance with your suggestion and resubmitted in a revised form for your consideration to be published in the BMC Infectious Disease.

We thank for the opportunity to improve the quality of our manuscript and have attempted most of the suggestions from you and reviewers into our revision. We hope the manuscript in its present revised form can meet your approval of publication.

Sincerely yours,

Yu-Chi Chiu, MD
Chest Department
Taipei Veterans General Hospital
#201, Section 2, Shi-Pai Road,
Taipei 112, Taiwan